Dear Lian-Sheng Ma,

Thank you for your helpful comments on my manuscript. We have made corrections according to the suggestions.

Reviewer (05449007):

Comments: Overall, well written retrospective study on bronchial trauma through blunt and penetrating injury. While the study is of sound nature, significant improvement in Grammar is recommended throughout the manuscript. Please mention the institutional board review (IRB) that approved the study. Recommend changing title to: "Evaluating management and outcome of bronchial trauma due to blunt versus penetrating Injuries" as it would provide the necessary information based on what the study tries to accomplish. The conclusion of the abstract and the manuscript need to be re-worded as it has to highlight the key differences identified in the study between blunt and penetrating injury.

Reply: A paragraph describing the issue of permission by the IRB was added in the method section. (Page 4, Line 70-82); the title has been changed to “Evaluating the management and outcome of bronchial trauma due to blunt versus penetrating Injuries”. We rewrite the conclusion part in the abstract and the main text, emphasizing the key difference between blunt and penetrating bronchial injury. (Page 3, Line 49-55; Page 14, Line 286-294)

Reviewer (05524138):

Comments: I agree with the authors that there is a problem of primary diagnosis of blunt trauma of the lungs and especially penetrating injuries of the bronchs. I will support the authors and recommend the publication.

Reply: Thanks for your comments.

Best regards,
Sincerely
Jinmou Gao
2021-12-04